Author's response to reviews

Title: Nasopharyngeal cancer mimicking Otitic Barotrauma in a resource challenged centre: a case report and literature review.

Authors:

Adekunle Daniel (kunle_d2002@yahoo.com)
Ayotunde J Fasunla (ayofasunla@yahoo.com)

Version: 3 Date: 4 August 2011

Author's response to reviews:

Dear Editor In-Chief,

JMCR

Dear Sir/Ma,


Thank you very much for the review of the manuscript and the comments. We have revised the manuscript as suggested and our comments are as follows:

1. Under case presentation the syntax 29 year old has been changed to 29-year-old.

2. In line 6 of the case presentation, “no associated nasal blockage, discharge, epistaxis or postnasal drip. No throat, neuro-ophthalmic symptoms or cervical swelling has been changed to: “she did not have nasal blockage, discharge, epistaxis or postnasal drip. There was no throat or neuro-ophthalmic symptom. She did not complain of neck swelling”.

3. In line 25 of the case presentation, the importance of Nasopharyngoscopy as an important aid in the diagnosis of nasopharyngeal cancer and the reason it was omitted in this index patient has been stated.

4. The importance of Nasopharyngoscopy has also been added to the discussion in line 23.

5. Other grammatical/syntax errors that were identified have been corrected.

Thank you once again.